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Device Service Record Form for JLab-2
	Date
	Comments

(Include the following in each DSR entry: What, Where, Why)
	Info Provided By

	5/15/2009
	email from Rongli Geng to CMG 4/13/2009:

J2 (one of the 9-cell built by JLab for FNAL) should be included. It reached 30.2 MV/m after the first light EP. This cavity has been since re-EP processed to explore the potential improvement. RF test is next week.
	Camille Ginsburg

	5/15/2009
	email from Jim Follkie to CMG (and others) 5/15/2009:

Currently in class-10 cleanroom awaiting final assembly for cold test.
	Camille Ginsburg

	6/11/2009
	Jim Follkie says this cavity is on the bench in the EP room at JLab.
	Camille Ginsburg

	6/23/2009
	Jim Follkie EP parameter spreadsheet, including JLAB-2 (J2) is here.
	Camille Ginsburg

	7/10/2009
	Jim Follkie: Cavity is in the EP room on a cart, to be moved to storage.
	Camille Ginsburg

	7/21/2009
8/21/2009

8/28/2009
	Jim Follkie: Cavity is in the Niobium Storage Area
	Camille Ginsburg

	9/8/2009
	This cavity was requested to be sent from JLab.  It will be used to commission one of the tuning machines from DESY currently in ICB.  It has the long-short cavity design.
	Camille Ginsburg

	9/15/2009
	Email from Jim Follkie:

I'd start looking for Jlab 2 today. It either shipped Friday or Monday.

Most likely Monday. I'll see if I can get tracking info before you get the

cavity.
	Camille Ginsburg

	10/15/09
	Cavity resides in ICB.  Cavity to be used for the commissioning of one of the DESY tuning machines in ICB.
	Jim Rife

	12/15/09
	Location verification.  Cavity in ICB 12/15/09.
	Jim Rife

	6/11/10
	Email from Rongli Geng 6/11/2010 regarding the second-pass test for this cavity:
JLab-2 last test reached 31.6 MV/m at Q0 1.2E10 limited by quench. The first-pass test was limited by Q-slope - but the second-pass successfully cured the Q-slope. 

The problem with JLab-2 is this is the few 9-cell cavities tested at JLab manifesting mode-mixing problem.
	Camille Ginsburg

	2010.06.15
	Cavity was optically inspected. Multiple defects on irises are noted. A pit on equator 8 is noted. Cavity is scheduled for molding of selected defects and tumbling after that.
	Dmitri Sergatskov

	16-Jun-2010
	Cavity was moved from ICB to IB4. The 4-bar frame was installed on it, and we will be making molds of the inner surface for defect analysis.
	Charlie Cooper

	6/17/2010
	Replica was done on Iris23 by Genfa Wu and Dave Burk
	Genfa Wu

	6/18/2010
	Replica was done on Iris67 by Genfa Wu and Dave Burk
	Genfa Wu

	22-June-2010
	Cavity tumbled for 5 hours with 3/8 inch ceramic oblong pyramid cutting media. Cavity rinsed but unable to remove plugs from input coupler and field probe side HOM can.
	Charlie Cooper

	1-July-2010
	Cavity removed from 4-bar frame and picked up by Brian Smith and Marco Battistoni to take to MP9 for ultrasonic degreasing.
	Charlie Cooper

	1-July-2010
	Cavity delivered to MP9 for ultrasonic degreasing.
	Charlie Cooper

	2-July-2010
	Cavity ultrasonically degreased by Marco Battistoni.
	Charlie Cooper

	6-July-2010
	Cavity moved from MP9 to optical inspection in ICB.
	Charlie Cooper

	6-July-2010
	Cavity optically inspected by Dmitri Sergatskov. Cavity waiting for 3 additional tumbling steps.
	Charlie Cooper

	7-july-2010
	Cavity molding applied to equator #8.
	MingQi Ge

	9-July-2010
	Cavity moved from ICB to IB4 to do final 3 tumbling steps.
	Charlie Cooper

	30-July-2010
	Cavity Finished Tumbling – received 5 tumbling stages total with one cutting media and 4 different polishing medias
	Charlie Cooper

	30-July-2010
	Horizontal Ultrasonic Degreasing in MP9 by Marco Battistoni.
	Charlie Cooper

	10-Aug-2010
	Optical Inspection Done in ICB my Jim Rife/Dmitri Sergatskov
	Charlie Cooper

	8/11/10
	Cavity moved from ICB to RF tuning area
	Damon Bice

	8/11/10
	Timer said he measured the cavity field flatness and is now done.  It is going next to ANL for rough-HPR, light-EP and then to JLab for 800C HT.
	Camille Ginsburg

	9/28/10
	Cavity transported to ANL, received HPR, 10 micron EP, and another HPR
	Damon Bice

	10/5/10
	Cavity transported to ICB for optical inspection
	Damon Bice

	08-Oct-2010
	Cavity is on optical inspection table.
	Dmitri Sergatskov

	10/11/10
	Cavity optically inspected, awaiting transport to Argonne for HPR then heat treatment at Fermi.
	Charlie Cooper

	14-Oct-2010
	Cavity is back on the optical inspection system for software development. 
Can be taken any time it is needed for transfer to Argonne.
	Dmitri Sergatskov

	15-Oct-2010
	Cavity is off the optical inspection, packed in the box, and ready for pickup.
	Dmitri Sergatskov

	10/25/2010
	Cavity moved from ICB to IB4 per Timergali K.
	RE / JB

	11/2/10
	Transported to ANL for U.S. cleaning and 1 pass HPR in prep for H Bake out
	Damon Bice

	11/4/10
	Transported to IB 4 for eventual furnace treatment
	Jim Follkie

	12/9/10
	Received 800 degree C, 180 minute furnace treatment
	Jim Follkie

	12/15/10
	JLAB2 cavity field flatness tuned, 99.2%, F=1297.44 MHz JLAB2 tuning. 2010.12.15.
	Timergali Khabiboulline

	1/4/11
	Cavity in crate at IB4
	Jim Follkie

	1/31/11
	Cavity transported to ANL for 10 um EP / HPR / Assy sequence
	Damon Bice

	2/1/11
	Cavity received 10um EP, HPR, Assembly at ANL
	Chris Baker

	2/8/2011
	Emails from Rongli Geng 2/8/2011 regarding additional vertical tests of this cavity:

The test mentioned in my original e-mail [see June 11, 2010 entry] was done on November 14, 2008. 

April 22, 2009 is last test of cavity JLab-2.
Final gradient 10 MV/m, final Q0 1.86E10. 
At this gradient 8/9-Pi appeared with Pi mode feeding, preventing correct measurement of the Pi-mode gradient and Q values.
This cavity had shown mode mixing (8/9-Pi excitation while feeding only Pi mode) several times over the course of testing. 
There is no preferred cell from T-mapping studies of the cavity.
	Camille Ginsburg

	3/15/11
	Vertical test result : Cavity showed early FE onset (~12 MV/m). Performed some pulse-processing at this gradient level, with minimal improvement. FE continued out to 16 MV/m (increasing in intensity), where additional pulse-processing was attempted At that time FE dramatically increased, leading to FE-loading of Q, beginning at about 12 MV/m, and leading to a Q of ~7E9 at ~16MV/m, where the cavity is now limited by a FE-induced quench. Second sound indicates the quench origin at cell #3. This was further confirmed by taking data at 1.75K. Data at 1.75K also showed earlier 

(9-10 MV/m) onset of Q-drop due to FE loading, indicating that the FE is getting progressively worse as the cavity is operated.
	J. Ozelis

	3/18/11
	Cavity is in its crate ready for pickup under vacuum still
	M carter

	3/22/11
	Cavity has been picked up and moved to ICB per A. Rowe
	JB/LZ

	      6/9/11
	VT 15-Mar.: heavy field emission, limitation 16 MV/m (FE/quench).  See Ginsburg Cavity Scheduling Mtg. 1-June-2011 slides for details.
	Jim Rife per Ginsburg Cavity Scheduling Mtg. 1-June-2011.

	      6/29/11
	VT 15-Mar.: heavy field emission, limitation 16 MV/m (FE/quench).  See Ginsburg Cavity Scheduling Mtg. 29-June-2011 slides for details.
	Jim Rife per Ginsburg Cavity Scheduling Mtg. 29-June-2011.

	      7/13/11
	VT 15-Mar.: heavy field emission, limitation 16 MV/m (FE/quench).  See Ginsburg Cavity Scheduling Mtg. 13-July-2011 slides for details.
	Jim Rife per Ginsburg Cavity Scheduling Mtg. 13-July-2011.

	8/21/12
	Cavity location verification.  Cavity @ ICB.
	Jim Rife
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